Author's response to reviews

Title: Seasonal variation of serum KL-6 levels is greater in patients with hypersensitivity pneumonitis

Authors:

Hiroshi Ohnishi (honi@kochi-u.ac.jp)
Shintaro Miyamoto (miyamoto@kochi-u.ac.jp)
Shigeo Kawase (shigeokawase@kochi-u.ac.jp)
Tetsuya Kubota (kubotat@kochi-u.ac.jp)
Akihito Yokoyama (ayokoyama@kochi-u.ac.jp)

Version: 2  Date: 25 April 2014

Author's response to reviews:

On behalf of all the authors, I would like to ask you to consider our manuscript entitled “Seasonal variation of serum KL-6 levels is greater in patients with hypersensitivity pneumonitis” for publication in the BMC Pulmonary Medicine as an original research article. This study compared seasonal variations of serum KL-6 levels in various interstitial lung diseases and found that greater seasonal variation in serum KL-6 levels was diagnostic for hypersensitivity pneumonitis. We feel that findings from this study will be of special interest to the readers of the BMC Pulmonary Medicine.

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission. This work was supported in part by Grants-in-Aid for Scientific Research of the Japanese Ministry of Education, Culture, Sports, Science, and Technology of Japan. The authors report no conflicts of interest.